Author’s response to reviews

Title: Microbicides Development Programme: Engaging the community in the standard of care debate in a vaginal microbicide trial in Mwanza, Tanzania

Authors:

Andrew J Vallely (a.vallely@uq.edu.au)
Charles Shagi (Charles.Shagi@Amref.org)
Shelley Lees (shelley.lees@lshtm.ac.uk)
Katherine Shapiro (shapiro.kathy@gmail.com)
Joseph Masanja (josephmasanja@yahoo.com)
Lawi Nikolau (josephmasanja@yahoo.com)
Johari Kazimoto (josephmasanja@yahoo.com)
Selephina Soteli (nyandashagi@gmail.com)
Claire Moffat (claire.moffat@lshtm.ac.uk)
John Changalucha (jchangalucha@yahoo.com)
Sheena McCormack (smc@ctu.mrc.ac.uk)
Richard J Hayes (richard.hayes@lshtm.ac.uk)

Version: 3 Date: 30 September 2009

Author’s response to reviews: see over
## Formatting changes requested

<table>
<thead>
<tr>
<th>Major revisions (We require the author to make these changes)</th>
<th>Action</th>
</tr>
</thead>
<tbody>
<tr>
<td>1. <strong>Author groups</strong> - If you wish to include the name of the group in the list of authors on the title page, you must remove the names of individual members of the group from the authors list in the submission system and the manuscript. You must also add the name of the group to the first name box in the submission system, and add the abbreviation of the name in brackets to the last name box. All the members of the group should then be listed in the Acknowledgments section. Alternatively, you may include the names of all authors from the group in the list of authors on the title page. If you do this, you must also include all the authors in the submission system. You should omit the name of the group from the list of authors on the title page and the submission system, but you should add a note in the Acknowledgements listing the members of the group, and describe the role of the group in this study in the authors' contributions section.</td>
<td>1. The original group authorship reference on the title page has been removed and the manuscript title modified to ‘Microbicides Development Programme: Engaging the community in the standard of care debate in a vaginal microbicidal trial in Mwanza, Tanzania’</td>
</tr>
<tr>
<td>2. <strong>Conclusions</strong> - Please insert a Conclusions section after the Discussion.</td>
<td>2. Completed</td>
</tr>
<tr>
<td>3. <strong>Reference 20, 21 and 22</strong> - Please provide author details</td>
<td>3. Completed</td>
</tr>
<tr>
<td>4. <strong>Box 1 and 2</strong> - Unfortunately due to the format of the on-line manuscript we cannot incorporate text boxes. Please include this information, in a plain text format, as appendices. An appendix section should appear before the Acknowledgements and all appendices should be cited in the following manner (Appendix 1) etc.</td>
<td>4. Boxes 1 &amp; 2 are now presented as Figures 2 &amp; 4 respectively in the revised manuscript</td>
</tr>
<tr>
<td>5. <strong>Table vertical elements</strong> - Please note that we are unable to display vertical lines within the tables nor are we able to correctly display merged cells where the merged cell crosses rows. Please re-layout your table without these.</td>
<td>5. All tables have now been reformatted</td>
</tr>
<tr>
<td>6. <strong>Tables as additional files</strong> - We notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript. This section should list the following for each file: the file format, the title of the data, and a short description of the data.</td>
<td>6. All tables are now included within the main manuscript file</td>
</tr>
<tr>
<td>7. <strong>Table shading</strong> - We are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold.</td>
<td>7. Shading removed</td>
</tr>
<tr>
<td>8. <strong>Figure 2</strong> - The image file should not include the figure legend. The legend and title should be part of the manuscript file after the reference list.</td>
<td>8. Completed</td>
</tr>
</tbody>
</table>
B. Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

1. Please remove the lines from the title page, around the title and from underneath every heading

2. Underlining - please note that underlining is not supported, you may wish to change this to bold

3. Word count - please remove this

4. Email address - please include email addresses for all the authors on the title page.

5. Highlighting - Please remove all highlighting

6. Authors' contributions - please ensure this is formatted into a paragraph rather than a list

7. <attached as separate files> - Please remove this text from below the figure legends and tables

8. Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.